Author's response to reviews

Title: An innovative OSCE Clinical Log station: a quantitative study of its influence on Log use by medical students

Authors:

Judith N Hudson (nicky.hudson@une.edu.au)
Helen Rienits (hrienits@uow.edu.au)
Linda Corrin (lcorrin@uow.edu.au)
Martin Olmos (molmos@uow.edu.au)

Version: 3 Date: 20 October 2012

Author's response to reviews: see over
October 29th, 2012

Editor,
BMC Medical Education

RE: MS: 1564897157768661

An innovative OSCE Clinical Log station: a quantitative study of its influence on Log use by medical students

Thank you for acceptance of the paper above.

- Initials have been given on page 8 as requested
- I don’t believe there is room in the paper to include the methodology for the evaluation in the methods section so I have removed the 2 paragraphs that you requested be removed, and referencing adjusted given that reference 28 was thus removed.

Professor Judith N (Nicky) Hudson